Author's response to reviews

Title: Encephalitis temporally associated with live attenuated Japanese encephalitis vaccine: Four case reports

Authors:

Na Jia (jiana79_41@hotmail.com)
Qiu Min Zhao (qmzhao@yahoo.com.cn)
Xiao Fang Guo (gxf@yipd.org)
Jing Xia Cheng (chengjingxia007@163.com)
Chao Wu (wuchao_simao@163.com)
Shu Qing Zuo (zuoshq@yahoo.com.cn)
Pei Fang Dai (prettynana@sohu.com)
Jun Ying Zhao (602044074@qq.com)
Jiu Song Zhang (zhang_jiusong@yahoo.com.cn)

Version: 6 Date: 9 November 2011

Author's response to reviews: see over
Dear Editor,

It is our pleasure to submit our revised manuscript entitled “Encephalitis temporally associated with live attenuated Japanese encephalitis vaccine: Four case reports”. We would be very grateful if it could be accepted for publication in your journal. Below you will find our responses to the reviewer’s comments.

1. Reply: We corrected it (Page 5 Line 11).
2. Reply: We corrected it (Page 6 Lines 18-19)
3. Reply: We corrected it (Page 7 Lines 1-2)
4. Reply: We corrected it (Table 1)
5. Reply: We corrected it (Table 1)

Best regards.

Jiu-Song Zhang, Dr.

Beijing Institute of Microbiology and Epidemiology, Beijing, China